

ANSWERING REVIEWERS

October 17, 2021

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 70231-review.doc).

Title: Pathophysiological mechanisms of hepatic stellate cells activation in liver fibrosis

Author: Dmitry Victorovich Garbuzenko

Name of Journal: *World Journal of Clinical Cases*

Manuscript NO.: 70231



1. Revision has been made according to the suggestions of the reviewer (number ID 05755592):

(1) The language has been edited.

(2) Text and Figure 1 have been added to the "INTRODUCTION" section. In addition, Figure 2 has been added to the "CONCLUSION" section.

The author would like to thank the reviewer for a positive conclusion regarding this manuscript.

Thank you again for publishing my manuscript in the *World Journal of Clinical Cases*.

Sincerely,

A handwritten signature in blue ink, appearing to read 'Dmitry Garbuzenko', with a large, stylized initial 'D'.

Dmitry Victorovich Garbuzenko, MD, PhD, Professor

Professor of Department of Faculty Surgery

South Ural State Medical University

Box 12317, Chelyabinsk

454092, Russia

E-mail: garb@inbox.ru